Author's response to reviews

Title: Study Protocol Acupuncture, or non-directive counselling versus usual care for the treatment of depression: a pilot study.

Authors:

Sylvia Schroer (ss51@york.ac.uk)
Hugh MacPherson (hm18@york.ac.uk)

Version: 2 Date: 7 November 2008

Author's response to reviews:

Have changed the title of the article, modified abbreviations, added competing interests statement and author’s contributions. The discussion has been amended so as to describe a significant association between poor attendance patterns and low rates of questionnaire return, and another table has been added.